RESEARCH

Reproductive Health

Open Access

Factors influencing the use of reproductive health services among young women in Nepal: analysis of the 2016 Nepal demographic and health survey



Naba Raj Thapa

Abstract

Background: Utilization of reproductive health services is a key component for preventing young women from different sexual and reproductive health problems. Thus, the objective of this study is to determine the factors influencing the use of reproductive health services among young women in Nepal.

Methods: Data have been extracted from the 2016 Nepal Demographic and Health Survey (NDHS) datasets wherein the weighted sample population size was restricted for modern contraceptive use to 1593 whereas for the antenatal care and skilled birth attendants to1606. This study has selected three reproductive health indicators as outcome variables of reproductive health service utilization for the analysis viz, modern contraceptive use, at least four antenatal care visits, and use of skilled birth attendants. Likewise, all calculations are based on standard sample weight of NDHS.

Results: The study has found that 21% of young women used modern contraception, 71% attended at least four ANC visits, and 67% utilized a skilled birth attendant at delivery. Young Janajati women, women having 1–2, and 3 or more living children, women participating household decision-making, and the ones having exposure to media were more likely to use modern contraceptives, whereas the women who want more children were less likely to use them. Higher education attainments, higher wealth quintile, and lower birth order were associated with higher level of receiving at least four ANC visits and SBAs. However, the young women willing to have more children and having access to media have higher odds of receiving at least four ANC visits; and the women attending four and more ANC visits have higher odds of using SBAs.

Conclusions: In order to improve the use of reproductive health services among young women, efforts should be made to enrich the young women of lower educational level, lower economic status, higher birth order, and lower exposure to media. Further research is required to detect the causes that affect the use of reproductive health services.

Keywords: Antenatal care, Contraceptive use, Reproductive health services, Skilled birth attendants, Young women

Correspondence: nabarthapa@gmail.com Department of Population Studies, Ratna Rajyalaxmi Campus, Tribhuvan University, Kathmandu, Nepal



[©] The Author(s). 2020 **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Plain English summary

Reproductive health is one of the most important components of human health, and poor reproductive health has more severe consequences in women. The studies and resources on factors associated with the use of reproductive health services among young women in Nepal are limited. Therefore, this study assesses the factors influencing the use of reproductive health services by young women in Nepal using the data from the 2016 Nepal Demographic and Health Survey. The study includes population of married women aged 15–24 years. The sample population for contraceptive use is 1593 and for antenatal care (ANC) visits and skilled birth attendant (SBA) is 1606.

About 21% of young women used modern contraceptive methods. Likewise, 71% of the young women attended at least four ANC visits and 67% used SBA.. Furthermore, the number of living children, fertility preference, women's participation household in decision-making and exposure to media were some important factors influencing modern contraceptive use. Besides, women's education, working status, wealth quintile, birth order, fertility preference and exposure to media were the main determiners of at least four ANC visits. Similarly, women's education, wealth quintile, birth order and ANC visits were the main factors influencing the use of SBAs.

From this perspective, efforts should be made, targeting the young women of lower educational level and economic status, lower exposure to media and higher birth order to increase the utilization of the reproductive health services in Nepal.

Background

In Nepal, about one-fifth of the total population comprises young people aged 15–24 years [1]. These young people are one of the important sub-groups of population from the concern of reproductive health. It is obvious that young people are facing diverse sexual and reproductive health challenges like unintended pregnancy, pregnancy related health care and sexually transmitted infections including HIV [2, 3]. Moreover, young women aged 15–24 years are at high risk of complications of pregnancy and death [4].

The reproductive health is the basic aspect of overall health [5] in both women and men. A broader concept of reproductive health focuses on contraceptive choices, maternal and newborn health cares, sexually transmitted infections and sexual health [6]. The increasing use of reproductive health services helps reduce maternal mortality and improve the health of children and women. To reduce global maternal mortality and ensure universal access to sexual and reproductive healthcare services are the targets of Sustainable Development Goal (SDG) 3 [7]. For achieving the targets of SDG 3, improving

women's sexual and reproductive healthcare services are essential.

The International Conference on Population and Development (ICPD) 1994 defines that reproductive healthcare is the constellation of methods, techniques and services that contribute to reproductive health and wellbeing by preventing and solving reproductive health problems [8]. The reproductive healthcare service includes family planning counselling, services for prenatal care, safe delivery and postnatal care, prevention and appropriate care related to reproduction and sexually transmitted diseases [5].

Before the ICPD held in Cairo in 1994, adolescent and youth specific reproductive health services were not existent in Nepal [9]. With the growing concern on reproductive health, the Government of Nepal developed the National Reproductive Health Strategy (1998) for providing integrated reproductive health services. Besides this, the National Adolescent Health and Development Strategy (2000) was also introduced to enhance the health and socioeconomic status of adolescent and youth.

The use of contraception can prevent unwanted pregnancy, early childbearing, and abortion related death. Although the knowledge of family planning is almost universal among adolescent and youth, the use of modern contraceptive is low among young women [10, 11]. Antenatal Care (ANC) is a key component to monitor and reduce the risk of mortality among young mothers during pregnancy and childbirth. Likewise, skilled birth attendance (SBA) is another important factor in reducing maternal and neonatal mortality. Nevertheless, only 21% of married young women aged 15–24 were found using modern contraceptives and about 72% were found receiving four or more ANC visits [12].

The government of Nepal has termed young women as a vulnerable and underserved group to accomplish the specific service required to meet their particular problems and needs. Independent studies conducted in Ethiopia [13, 14], Bangladesh [15, 16] and Nepal [17] have revealed that reproductive health service utilization among adolescent and youth was low and the use of reproductive health services was associated with different socio-demographic and economic factors. In Nepal, there are limited studies regarding the use of reproductive health services among young women. In this connection, this study was carried out to determine the factors influencing the use of reproductive health services among young women in Nepal.

Data and methods

Data sources

This study has utilized the data from the 2016 Nepal Demographic and Health Survey (NDHS) datasets. The NDHS is a nationally representative cross-sectional survey conducted under the aegis of the Ministry of Health of the Government of Nepal. In the 2016 NDHS, 11473 households were selected as samples of which 11,040 were interviewed. Similarly, from among 13,089 women aged 15-49 identified for individual interviews, only 12, 862 women were successfully interviewed with the response rate of 98%. To select the household for the survey, stratified two-stage cluster sampling was used in rural areas whereas three-stage in urban ones. In rural areas, wards were selected as primary sampling units (PSUs) in the first stage and the households were finalized in the second stage. In urban areas, wards were selected as PSUs in the first stage, one enumeration area (EA) was selected from each PSU in the second stage, and households were selected from sample EAs in the third stage. Next, face-to-face interviews of eligible women aged 15-49 in the sample households were conducted with structured questionnaires.

There were six questionnaires administered in the 2016 NDHS: the household questionnaire, the women's questionnaire, the men's questionnaire, the biomarker questionnaire, the fieldworker questionnaire and the verbal autopsy questionnaire. This study used the data collected through women's questionnaire which was utilized to collect information from among the married women aged 15-49 years. The women's questionnaire sought the information of background characteristics, reproductive history and child mortality, knowledge and use of family planning methods, fertility preferences, delivery care, child health, women's works, antenatal delivand postnatal care, husband's background ery characteristics and domestic violence [18]. The Individual Record (IR) data file was employed for this study. We confined the analyses to the 1593 young women who were currently married and non-pregnant for modern contraceptive use and 1606 young women who had birth in the 5 years preceding the survey for ANC visits and SBA for their most recent birth.

Variables

Dependent variables

Here, three reproductive health indicators were selected as dependent variables for the analysis. The first dependent variable is the current use of modern contraception which is dichotomous. Young women who were currently using modern contraception were coded as 1 and otherwise as 0. The modern contraceptive methods include male and female sterilization, injectables, intrauterine devices, pills, implants, male condoms, locational amenorrhea, and emergency contraceptives.

The second dependent variable is ANC visits for the most recent pregnancy in the 5 years preceding the survey, which is also dichotomous. The number of ANC

visits were coded as 1 if women attended four or more ANC visits before their most recent birth and coded as 0 if there were fewer than four visits. Then the third dependent variable is the assistance of the SBA which is also dichotomous. Here, women assisted by the SBA during most recent delivery were coded as 1 and otherwise as 0. The SBAs include doctors, nurses, and mid-wives [18, 19].

Independent variables

The independent variables of this study are women's age, education, ethnicity, occupation, wealth quintile, birth order, number of living children, participation in household decision-making and exposure to media. The age groups of women in this study are 15-19 and 20-24. Mother's age at birth was classified into two age groups- < 20 and 20-24. Education level achieved by young women was classified into four categories: no education, primary, some secondary, and SLC and above. Moreover, caste/ethnicity variable was categorized into five groups- Brahman/Chhetri, Terai caste, Dalit, Janajati, and Muslim/other. Women's working status was grouped into two categories- not working, and working. In the same way, wealth quintile variable was recorded into five categories- poorest, poor, middle, richer, and richest. Wealth quintile is a composite measure of household economic status composed of key household assets. Household wealth quintile was constructed using data of household assets such as televisions, bicycles, materials used for house construction, sources of drinking water, sanitation facilities and other wealth related characteristics [20]. Birth order was measured in three categories- 1, 2, and 3 or more. Similarly, the number of living children was measured in three categories- 0, 1-2,and 3 or more whereas fertility preference of young women was categorized into three groups- wants no more children, wants more children, and undecided. Women's participation in household decision-making was studied in terms of three decisions: women's own healthcare, large household purchases, and visit to family or relatives. Each of these three aspects of decisionmaking had five response options: respondent alone, respondent and husband/partner jointly, husband/partner, someone else and other. These responses were coded as 1 for yes (respondent alone, and respondent and husband/partner jointly) and 0 for no (husband/partner, someone else, and other). Then, a single composite variable was constructed by grouping women into two categories: women who participated in at least one decision were categorized as participated one or more decisions (=1) otherwise as not participated (=0). Media exposure is dichotomous variable. Finally, women exposed to either radio, television or newspapers/magazines at least

once a week were coded as exposed (=1) and those exposed to none as not exposed (=0).

Methods of data analysis

This study employed three levels of data analyses. Firstly, univariate analysis was carried out to analyze the selected socio-demographic characteristic of young women aged 15-24. Secondly, the chi-square test was carried out to examine the association between the modern contraceptive use, at least four ANC visits and the use of SBA, and explanatory variables such as sociodemographic, economic, women's participation in household decision-making and media exposure. Third, multivariate analysis with logistic regression was performed to examine the factors influencing the use of reproductive health services. Likewise, factors significantly associated with outcome variables at p < 0.05 were included in multivariate logistic regression. Multicollinearity assessment was employed before the multivariate analysis. The results were shown in adjusted odds ratios (OR) with 95% confidence interval (CI). The data were analyzed by using STATA version 15.1. All calculations were weighted using standard sample weight of NDHS. The STATA survey command 'svy' was applied to adjust for stratified sample design.

Results

Background characteristics of young women

Table 1 shows that more than half of young women were aged 15–19. Forty percent of young women completed the secondary level of education. About 43% of young women were not working but 57% were working. About 21% of young women lived in Province 2 and 20% lived in Bagmati Province. Janajati (36%) and Brahman/Chhetri (29%) were dominant castes/ethnicities. Nearly 18% of young women were from the poorest, 21% from the middle, and 18% from the richest wealth quintile. 32% young women had 1–2 living children and 68% had exposure to media. Overall, the majority of young women were aged 15–19, educated, working, Janajati and Brahman/Chhetri and exposed to media.

Bivariate analysis

Table 2 presents the relationship between modern contraceptive use and selected background characteristics. Here, less than one quarter of young women were found to have used modern contraception. The results revealed that the independent variables were significantly associated with modern contraceptive use at p < 0.05. Modern contraceptive use was positively associated with age group. The use of contraception was higher among women aged 20–24 than among women aged 15–19. In this study, young women's education and wealth quintile were insignificantly related to modern contraceptive use.

 Table 1
 Percent distribution of young women aged 15–24 by background characteristics, Nepal DHS 2016

Background characteristics	Percent	Number
Age		
15–19	53.6	2598
20–24	46.4	2251
Education level		
No education	10.0	483
Primary	14.3	696
Secondary	40.3	1953
SLC & above	35.4	1718
Working status		
Not working	42.7	2070
Working	57.4	2779
Province		
Province 1	16.8	817
Province 2	20.5	996
Bagmati	19.6	952
Gandaki	9.5	460
Province 5	18.4	892
Karnali	6.0	293
Sudurpashchim	9.1	440
Ethnicity		
Brahman/Chhetri	29.3	1420
Terai caste	15.2	737
Dalit	13.2	641
Janajati	36.2	1755
Muslim	6.1	296
Wealth quintile		
Poorest	17.5	847
Poorer	20.5	994
Middle	20.9	1015
Richer	22.8	1104
Richest	18.3	890
Number of living children		
0	66.0	3202
1–2	31.8	1543
3+	2.2	104
Exposure to media (at least one	ce a week)	
Not exposed	31.9	1548
Exposed	68.1	3301
Total	100.0	4849

Young women, who were better educated, working, Janajati, wealthier and having three or more living children; who wanted no more children, participated household decision making and had exposure to media had
 Table 2 Percentage of young women aged 15–24 using modern contraception according to background characteristics, Nepal DHS 2016

	Modern contraceptive use					
Background characteristics	Percent	95% CI	Number	p -value (χ²)		
Age						
15–19	14.5	[11.5–18.1]	704	0.000		
20–24	23.9	[21.3–26.8]	1684			
Education						
No education	15.9	[12.1-20.6]	396	0.085		
Primary	20.4	[16.4-25.0]	486			
Secondary	23.0	[19.6–26.7]	914			
SLC & above	22.4	[18.5–26.7]	593			
Working status						
Not working	17.8	[14.8-21.2]	1041	0.006		
Working	23.7	[21.0-26.7]	1347			
Caste/ethnicity						
Brahman/Chhetri	22.4	[18.8–26.5]	608	0.000		
Terai caste	13.5	[9.6–18.7]	463			
Dalit	17.4	[12.9–23.0]	366			
Janajati	28.6	[24.7–33.0]	795			
Muslim	9.3	[5.4–15.6]	157			
Wealth quintile						
Poorest	22.5	[18.8–26.7]	440	0.422		
Poorer	22.8	[18.4–27.9]	526			
Middle	18.7	[15.1-22.9]	589			
Richer	19.6	[15.6-24.4]	548			
Richest	23.9	[18.2–30.8]	286			
Number of living children						
0	8.5	[6.3–11.4]	754	0.000		
1–2	26.0	[23.2–29.0]	1532			
3+	42.0	[30.4–54.6]	103			
Fertility preference						
Wants no more children	33.6	[29.7–37.6]	745	0.000		
Wants more children	15.1	[12.9–17.8]	1540			
Undecided	20.9	[13.4–31.0]	103			
Women's participation in decision-making						
Not participated	16.2	[13.9–18.8]	1226	0.000		
Participated one or more decisions	26.4	[23.2–29.7]	1162			
Exposure to media (at least once a week)						
Not exposed	17.7	[14.8–21.0]	946	0.005		
Exposed	23.4	[20.7–26.4]	1443			
Total	21.1	[18.9–23.5]	2389			

the highest level of contraceptive use. Moreover, use of modern contraception was the lowest among young women who didn't have education and exposure to media; were in the middle wealth quintile, Muslim and not working; and did not participate household decision-making.

The associations between ANC visits and skilled birth attendant for most recent delivery according to selected background characteristics have been presented in Table 3. About 72% of young women made at least four ANC visits for the most recent birth. The bivariate analysis demonstrated that the independent variables, excluding age of women, were significantly associated (p < 0.05) with at least four ANC visits. The Mothers aged below 20 years at birth appeared more likely to make at least four ANC visits (73%). ANC visits varied as per women's education. 54% young women with no education made at least four ANC visits compared with 87% young women who attained SLC and above education. A large proportion of young women who are working (78%) reported at least four ANC visits. Besides, there is significant relationship between caste/ethnicity and ANC visits. A greater proportion of young Brahman/Chhetri women made at least four ANC visits while young Muslim women had a small proportion. Young women in the richest wealth quintile (80%) were more likely to make at least four ANC visits than those in the poorest wealth quintile (65%). The percentage of young women making at least four ANC visits decreased as birth order increased. Level of ANC visits was higher among those who wanted more children (77%), participated in household decision making (75%), and had exposure to media (79%).

Table 3 reflects the use of SBA for the most recent birth according to selected background variables. About 69% of young women used SBA in delivery. It was found that all background variables except mother's age at birth and working status had a statistically significant association (p < 0.05) with SBA. Mothers aged below 20 years at birth used SBA more often than those aged 20-24 years. The percentage of young women using SBA increased with the increase in women's education level from 52% among young women with no education to 80% among young women with SLC and higher education. A large proportion of young Brahman/Chhetri women (72%) used SBA. It was found that the use of SBA was positively related to the wealth quintile and negatively related to the birth order. Young women who were desirous of more children, participated in household decision making, and had exposure to media were more likely to use SBA.

Multivariate analysis

Table 4 presents the factors correlated with the use of modern contraception among currently married young women aged 15–24. In the multivariate analysis, women's education and wealth quintile were excluded because they were not found to be significant associations with modern contraceptives use in bivariate analysis. The results of multivariate logistic regression demonstrate that young women who were Janajati,

having 1-2 and 3 or more living children, participated one or more household decisions, and had exposure to media were more likely to use modern contraception. The young Janajati women were found 1.76 times more likely to use modern contraception than young Dalit women. The odds of modern contraceptives use were 3.22 times and 9.07 times greater for women with 1-2, and 3 or more children respectively compared to young women having no living children. The young women who wanted more children were less likely to use modern contraception (OR = 0.57, p < 0.001) compared with young women who wanted no more children. The odds of modern contraceptive use were 1.34 times greater among young women who participated one or more household decisions and 1.49 times greater among young women who had exposure to media compared with young women who did not participate in household decisions and had no exposure to media respectively.

Table 5 illustrates those factors associated with at least four ANC visits and SBA at delivery for most recent birth among young married women aged 15-24. Mother's age at birth was excluded from multivariate analysis because this variable was insignificantly associated with ANC visits. The results of logistic regression show that education (secondary, and SLC and above), working status, wealth quintile (middle, richer and richest), second birth order, young women who wanted more children, and exposure to media had significant association with attending at least 4 ANC visits. The analysis indicates that the odds of attending at least four ANC visits were 1.93 times greater among young women with secondary education compared with young women with no education; and 2.91 times greater among young women with SLC and higher education. The young working women were 1.90 times more likely to make at least four ANC visits, compared with nonworking ones. The young women in the richer wealth quintile were 2.55 times more likely to attend at least four ANC visits than young women in the poorest wealth quintile whereas young women in the richest and the middle wealth quintile appeared 2.12 and 1.85 times more likely to make at least four ANC visits than the women in the poorest wealth quintile respectively. The odds of receiving ANC visits were 36% lower for women with secondorder birth compared to women with first-order birth. The young women wanting more children were 1.65 times more likely to make at least four ANC visits compared to young women expecting no more children. The young women with exposure to media were 48% more likely to receive at least four ANC visits compared to those with no exposure.

Table 5 also depicts the results of the logistic regression of young women's use of SBA during delivery for most recent birth. Mother's age at birth and women's

Table 3 Percentage of young women	aged 15-24 who had at least fou	ur visits for ANC and SBA for r	most recent birth in the 5 years
preceding the survey according to bac	kground characteristics, Nepal Dł	-IS 2016	

	4+ ANC visits			Skilled birth attendant			
Background characteristics	%	95% CI	p -value (χ²)	%	95% CI	p -value (χ²)	Number
Mother's age at birth							
< 20	73.2	[68.7–77.3]	0.268	68.6	[64.0-72.8]	0.177	792
20–25	70.2	[65.7–74.4]		65.0	[60.8–69.0]		813
Education							
No education	54.1	[45.7–62.2]	0.000	52.2	[44.3–60.0]	0.000	307
Primary	63.0	[56.5–69.0]		56.6	[49.8–63.2]		350
Secondary	76.9	[72.4–80.9]		72.5	[67.7–76.9]		599
SLC & above	86.9	[82.9–90.1]		79.9	[74.9-84.1]		349
Working status							
Not working	65.3	[59.7–70.6]	0.000	69.3	[64.7–73.5]	0.104	707
Working	76.7	[73.1-80.0]		64.8	[60.5–68.9]		899
Caste/ethnicity							
Brahman/Chhetri	82.2	[77.6-86.0]	0.000	72.3	[66.1–77.8]	0.033	399
Terai caste	63.6	[53.5–72.6]		62.9	[55.4–69.8]		334
Dalit	67.2	[59.4–74.1]		57.9	[51.1-64.3]		241
Janajati	73.3	[67.8–78.3]		68.2	[61.7–74.1]		517
Muslim	60.7	[48.6–71.6]		70.9	[59.7–80.0]		114
Wealth quintile							
Poorest	64.6	[57.9–70.8]	0.007	45.5	[39.0–52.1]	0.000	314
Poorer	70.1	[64.1-75.5]		58.5	[52.1-64.6]		371
Middle	69.9	[62.9–76.0]		73.6	[67.7–78.7]		397
Richer	77.6	[71.4-82.8]		76.2	[70.7-80.9]		355
Richest	80.2	[71.5-86.8]		88.9	[80.7–93.9]		169
Birth order							
1	78.7	[75.3-81.8]	0.000	74.7	[71.1–78.1]	0.000	1034
2	61.3	[55.1–67.2]		55.8	[49.6–61.8]		434
3+	51.6	[42.1–61.0]		41.5	[31.8–51.8]		138
Fertility preference							
Wants no more children	64.1	[59.1–68.7]	0.000	62.8	[58.1–67.2]	0.005	685
Wants more children	76.9	[73.0-80.5]		69.1	[65.1–72.9]		854
Undecided	82.7	[70.2–90.6]		77.8	[65.9–86.3]		67
Women's participation in de	cision-making	I					
No	68.7	[64.1–73.0]	0.021	64.0	[59.6–68.2]	0.045	783
One or more decisions	74.6	[70.4–78.4]		69.3	[65.0-73.4]		810
Exposure to media (at least	once a week)						
Not exposed	62.5	[56.7–68.0]	0.000	56.9	[52.2–61.4]	0.000	687
Exposed	78.5	[75.0-81.7]		74.2	[69.9–78.1]		919
Total	71.7	[68.1–75.0]		66.8	[63.3–70.1]		1606

working status were excluded from multivariate logistic analysis due to insignificant association with SBA. The analysis shows that the odds of receiving SBA were 1.59 times greater among young women with secondary education compared to young women with no education and 1.35 times greater among young women with SLC

Background variable	Odds ratio	95% confidence interval
Age		
15–19	1.00	
20-24	0.98	0.71-1.34
Working status		
Not working	1.00	
Working	1.12	0.85–1.48
Caste/ethnicity		
Dalit	1.00	
Brahman/Chhetri	1.30	0.84-2.02
Terai caste	0.64	0.36–1.16
Janajati	1.76**	1.15–2.70
Muslim	0.52	0.25-1.05
Number of living children		
0	1.00	
1–2	3.22***	2.31–4.49
3+	9.07***	4.88–16.86
Fertility preference		
Wants no more children	1.00	
Wants more children	0.57***	0.45–0.73
Undecided	0.71	0.41–1.24
Women's participation in decision-making		
Not participate	1.00	
Participate one or more decisions	1.34*	1.04–1.74
Exposure to media (at least once a week)		
Not exposed	1.00	
Exposed	1.49**	1.15–1.92

Table 4 Odds ratios from logistic regression showing factors associated with the use of modern contraception among young women aged 15–24, Nepal DHS 2016

Note: **** *p* < 0.001, ** *p* < 0.01, * *p* < 0.05

and higher education. The odds of SBA increase as the level of wealth quintile rises. The young women in the richest wealth quintile had 9.16 times higher odds of receiving SBA compared to young women in the poorest wealth quintile. Birth order seemed to be another one of the most important predictors of SBA. Higher order births had lower odds of SBA. The odds of receiving SBA were 65% lower for three and more birth orders compared to first order birth and 51% lower for second order birth. The young women making at least four ANC visits had 2.59 times higher odds of receiving SBA compared to those making less than four ANC visits.

Discussion

The use of modern contraception (21%) by young women was lower than currently married women in the age of 15–49 years (43%). Among young women, the proportion of those receiving at least four ANC visits (72%) was higher compared to the ones in the 15–49 age

group (69%). Similarly, the proportion of young women assisted by SBA for the most recent birth (69%) was higher than the national average (61%).

This study shows that inequalities in contraceptive use among young women occurred in terms of caste/ethnicity. The young Jananati women had higher odds of modern contraceptive use compared to young Dalit women. Studies conducted in Nepal revealed that odds of using modern contraceptive were higher for Janajati women compared to other caste/ethnic groups [12, 21]. It is argued that differences in religious faith, and cultural belief and values between caste/ethnic groups may be some possible reasons for disparities in the use of contraception. There is a significant positive association between the number of living children and the use of modern contraceptives. Young women with more living children were more likely to use modern contraceptives that those with less living children. This finding is in line with the previous studies from Ghana [22], Malawi [23],

Table 5 Odds ratios from logistic regression showing factors associated with ANC visit (4 + ANC visits) and SBA among young women aged 15–24, Nepal DHS 2016

	4+ ANC Visit		Skilled birth attendant	
Background variable	Odds ratios	95% CI	Odds ratio	95% CI
Education				
No education	1.00		1.00	
Primary	1.25	0.85-1.83	1.14	0.77-1.68
Secondary	1.93***	1.31-2.84	1.59*	1.03–2.44
SLC & above	2.91***	1.79–4.76	1.35	0.84–2.16
Working status				
Not working	1.00			
Working	1.90***	1.43–2.53		
Caste/ethnicity				
Dalit	1.00		1.00	
Brahman/Chhetri	1.29	0.81-2.07	1.22	0.77-1.93
Terai caste	0.78	0.45-1.36	0.76	0.45-1.28
Janajati	0.83	0.53-1.31	1.04	0.73–1.48
Muslim	0.69	0.37-1.29	1.18	0.59–2.37
Wealth quintile				
Poorest	1.00		1.00	
Poorer	1.30	0.88-1.91	1.69**	1.19–2.40
Middle	1.85*	1.16-2.97	4.46***	2.70-7.38
Richer	2.55***	1.54-4.21	4.01***	2.56-6.27
Richest	2.12*	1.14-3.93	9.16***	4.47-18.76
Birth order				
1	1.00		1.00	
2	0.64**	0.46-0.88	0.49***	0.35-0.67
3+	0.70	0.45-1.08	0.35***	0.21-0.58
Fertility preference				
Wants no more children	1.00		1.00	
Wants more children	1.65***	1.25-2.18	0.86	0.65-1.12
Undecided	1.89	0.92-3.89	1.01	0.50-2.07
Women's participation in decision-making	3			
Not participate	1.00		1.00	
Participate one or more decisions	1.12	0.87-1.44	1.16	0.89–1.52
Exposure to media (at least once a week)				
Not exposed	1.00		1.00	
Exposed	1.48*	1.09-2.02	1.21	0.90-1.63
ANC visits				
Less than four visits			1.00	
Four or more visits			2.59***	1.95–3.43

Note: *** *p* < 0.001, ** *p* < 0.01, * *p* < 0.05

and Ethiopia [24]. This may be argued that women with more children may reach the desired family size so that they are more likely to use contraceptives than women with no or less living children. Fertility preference has been identified as an important factor to influence the use of modern contraceptives. That is, young women who want more children are less likely to use modern contraceptives than women who want no more children. This finding is consistent with different studies conducted in Ethiopia [25], Bangladesh [26], Nepal [11, 21] and Nigeria [27]. This study has also found that young women who participate in household decision making have higher odds of using modern contraceptives than the ones not participating in such decision making process. This finding is consistent with previous studies that have shown a strong association between women's participation in household decision making and use of modern contraceptives [24, 27, 28]. Such a situation might take place because educated and employed women are primarily involved in household decision making and may have more liberty to use contraceptives.

Exposure to media was found to have a positive association with modern contraceptive use [22, 29, 30]. Television, radio and newspapers were most commonly used sources of information regarding contraceptives in Nepal [31]. This study also indicates that young women exposed to media are more likely to use modern contraceptives. The reason behind this may be the women with access to mass media keep good knowledge on reproductive health problems and services which helps them in improving their knowledge of and build confidence towards using modern contraceptives.

The results from this study have revealed that higher educational level of mothers, higher wealth quintile, and exposure to media are positively associated with at least four ANC visits while higher birth order is negatively associated. These findings are consistent with studies conducted in Indonesia [32], India [33], and Nepal [11]. The study also found that young women who want more children are more likely to receive ANC than the ones who do not. This finding is consistent with the study undertaken in Benin [34]. Women's working status is found to be a significant predictor of using ANC as young working women are more likely to attend at least four ANC visits than not working ones. This finding is supported by a previous study conducted in Ethiopia [35].

Our finding suggests that the women's education level is an important variable affecting the use of SBA because young women with secondary and higher education are more likely to get assisted by SBAs during delivery. This finding is consistent with previous studies conducted in Nepal [36] and Ghana [37]. Likewise, this study has found that the use of SBA is inequitable in Nepal. Women with low wealth quintile are less likely to use SBAs. This is in line with the result of previous studies conducted in the South Asian Region [38] and Bangladesh [39]. Also, the finding on birth order and SBAs is consistent with the result of a study conducted in the South Asian Region [38] and India [40], as young women with high birth order are less likely to seek SBAs. The finding of this study revealed that the use of ANC is strongly associated with using SBA among young women. Young women who attend at least four ANC visits are more likely to have SBAs in delivery. This findings is consistent with previous studies in Nepal [36], and Ghana [37].

The possible explanation of this relationship can indicate that educated women may have acquired more knowledge about reproductive health and better understanding of benefits of ANC and SBAs; as a result, they have a higher level of ANC visits and a higher chance of having SBAs at delivery. Women in the richest wealth quintile are more likely to be more educated and have the financial capacity to afford the ANC visits and to seek the help of the SBAs in delivery. But mothers of first-order birth have no experience of childbirth, they fear from the possible complication of childbirth. Thus, they are more likely to receive ANC and use SBAs. Meanwhile, this practice has declined subsequent births due to the experience of pregnancy and childbirth. Exposure to media may increase women's knowledge or awareness and positive attitude towards ANC and SBAs. As a result, young women who read newspapers, listen to the radio and watch television are more likely to use reproductive health services. In relation to women's working status, those who are working are more likely to be educated, involve in household decision making and get access to reproductive health services than those who are not. The ANC visits help young women have better understanding of risk in regard to the complication of pregnancy and childbirth and enable them to make a decision regarding the use of SBAs [41].

Limitations and strengths

This study is based on the data of a cross-sectional survey from the 2016 NDHS. Therefore, this study is limited to establishing causal relationships between variables. However, efforts were made to examine the association between explanatory variables and outcome variables. This study has included those variables which were available for analysis of DHS. Another limitation is that this study does not include time series data in the analysis. The main strength of the study is the use of nationally representative data and multivariate methods to identify factors influencing the use of reproductive health services among young women.

Conclusions

To sum up, there has been significant improvement in reproductive health indicators among young women. The finding of this study has revealed that number of living children, fertility preference, women's participation in household decision-making and exposure to media are the important determiners of modern contraceptive use among young women in Nepal. Therefore, family planning programs should focus on young women of low parity and those who want more children. This study has also identified women's education, working status, wealth quintile, birth order, fertility preference and exposure to media as factors influencing of ANC visits. Likewise, the study has shown that women's education, wealth quintile, birth order, and ANC visits significantly influence the use of SBAs in delivery among young women. Efforts to encourage attending at least four ANC visits and taking assistance of SBAs should be targeted at young women having lower educational and economic status. The study has also indicated that improving women's education level, generating employment opportunities, improving economic status, proper media campaigns and encouraging them to attend at least four ANC visits are important steps to increase the use of reproductive health services in Nepal. In order to expand the use of reproductive health services among young women, necessary efforts should be made targeting young women of lower educational level, economic status and exposure to media, and higher birth order. Finally, the results of the study have indicated that further research is required to understand the perspective of young women towards reproductive health service thereby exploring the barriers of the use of reproductive health services.

Abbreviations

ANC: Antenatal Care; Cl: Confidence Interval; DHS: Demographic and Health Survey; EA: Enumeration Area; NDHS: Nepal Demographic and Health Survey; OR: Odds Ratio; PSU: Primary Sampling Unit; SBA: Skilled Birth Attendant; SDG: Sustainable Development Goal; SLC: School Leaving Certificate

Acknowledgements

I am grateful to the DHS Program for making Nepal Demographic and Health Surveys (NDHS) datasets accessible for this study. I also fondly thank Dr. Wenjuan Wang and Dr. Shireen Assaf, who made it possible to write this paper with appropriate use of DHS data. My special gratitude goes to Dr. Badri Bhattarai, Old Dominion University, Virginia and Dr. Ripu Kunwar for helping to edit the manuscript. I am also thankful to reviewers for valuable comments and intensive review of the paper.

Author's contributions

Naba Raj Thapa conducted data analysis, reviewed literature, made interpretation and drafted the paper. The author(s) read and approved the final manuscript.

Authors' information

Naba Raj Thapa holds a Masters of Philosophy (M.Phil.) in Population Studies from the Central Department of Population Studies (CDPS), Tribhuvan University, Nepal. He is currently working as Associate Professor at the Department of Population Studies, Ratna Rajyalaxmi Campus, Tribhuvan University, Nepal.

Funding

The author received no funding for this study.

Availability of data and materials

The data used in this study are freely available from The DHS Program website (https://www.dhsprogram.com/data/available-datasets.cfm). The study was carried out specifically using 2016 Nepal Demographic and Health Survey dataset.

Ethics approval and consent to participate

Data from the 2016 Nepal Demographic and Health Survey were used for this study. The DHS Program granted access to the data after reviewing the

research proposal. The ethical approval for the survey was obtained by ICF International Institutional Review Board (IRB). Further approval for this study was not required since data are freely available in the public domain.

Consent for publication

This study does not include data from any person. Therefore, consent to publish is not applicable.

Competing interests

The author has no competing interests.

Received: 5 February 2020 Accepted: 18 June 2020 Published online: 29 June 2020

References

- 1. Central Bureau of Statistics. National Population and housing census 2011, national report. Kathmandu: Central Bureau of Statistics; 2012.
- Regmi PR, Teijlingen EV, Simkhada P, et al. Barriers to sexual health services for young people in Nepal. J Health Popul Nutr. 2010;28:619–27.
- Fatusi AO. Young people's sexual and reproductive health interventions in developing countries: Making the investments count. J Adolesc Health; 59. Epub ahead of print. 2016. https://doi.org/10.1016/j.jadohealth.2016.06.016.
- World Health Organization. Maternal Mortality. 2019, https://www.who.int/ news-room/fact-sheets/detail/maternal-mortality (Accessed 19 Sept 2019).
- United Nations Population Fund. The State of World Population 1997. The right to choose: reproductive rights and reproductive health. New York: United Nations Population Fund; 1997.
- United Nations Population Fund. Sexual and Reproductive Health for All. New York: United Nations Population Fund. http://www.unfpa.org/public/ home/publications/pid/6526; 2016.
- National Planning Commission. Nepal's Sustainable Development Goals Status and Roadmap: 2016-2030. Kathmandu: Government of Nepal, National Planning Commission; 2017.
- Roudi-Fahimi F, Ashford L. Sexual and reproductive health in the Middle East and North Africa: a guide for reporters. Washington, DC: Population Reference Bureau; 2008.
- Ministry of Health. National Adolescent Health and Development Strategy. Kathmandu: Ministry of Health [Nepa]], Department of Health Services; 2000.
- 10. Khan S, Mishra V. Youth reproductive and sexual health. DHS comparative reports no. 19. Calverton: Macro International Inc.; 2008.
- Khatiwada N, Silwal PR, Bhadra R, et al. Sexual and Reproductive Health of Adolescents and Youth in Nepal: Trends and Determinants: Further analysis of the 2011 Nepal Demographic and Health Survey.
- Kafle RB, Paudel R, Gartoulla P, et al. Youth Health in Nepal: Levels, Trends, and Determinants. DHS Further Analysis Report No. 116. Rockville: ICF; 2019. https://dhsprogram.com/pubs/pdf/FA116/FA116.pdf.
- Tlaye KG, Belete MA, Demelew TM, et al. Reproductive health services utilization and its associated factors among adolescents in Debre Berhan town, Central Ethiopia: a community-based cross-sectional study. Reprod Health. 2018;15:1–11.
- Negash W, Dessalegn M, Yitayew B, et al. Reproductive health service utilization and associated factors: the case of north Shewa zone youth, Amhara region, Ethiopia. Pan Afr Med J. 2016;25:3.
- Shahabuddin ASM, Delvaux T, Abouchadi S, et al. Utilization of maternal health services among adolescent women in Bangladesh: a scoping review of the literature. Trop Med Int Health. 2015;20:822–9.
- Haque SE, Rahman M, Mostofa MG, et al. Reproductive health care utilization among young mothers in Bangladesh: does autonomy matter? Womens Health Issues. 2012;22:e171–80.
- Subedi G, Dwivedy RK. Adolescent Girls Reproductive Health Situation in Nepal : A Case Study from Mahottari District, http://library.nhrc.gov.np:8080/ nhrc/handle/123456789/608 (2009).
- Ministry of Health N, New ERA, ICF. Nepal Demographic and Health Survey 2016. Kathmandu: Ministry of Health, Nepal; 2017.
- World Health Organization. Sexual and reproductive health: Skilled birth attendants, https://www.who.int/reproductivehealth/publications/maternal_ perinatal_health/9241591692/en/ (2019).
- 20. Rutstein SO, Kiersten J. The DHS wealth index. DHS comparative reports no. 6. Calverton: ORC Macro; 2004.
- Lamichhane KD. Use of contraceptive methods among young married women in Nepal. Tribhuvan Univ J. 2018;32:75–96.

- Aviisah PA, Dery S, Atsu BK, et al. Modern contraceptive use among women of reproductive age in Ghana: analysis of the 2003-2014 Ghana demographic and health surveys. BMC Womens Health. 2018;18:1–10.
- Palamuleni ME. Socio-economic and demographic factors affecting contraceptive use in Malawi. Afr J Reprod Health. 2013;17:91–104.
- Wado YD. Women's autonomy and reproductive healthcare-seeking behavior in Ethiopia. DHS working papers no. 191. Calverton: ICF International; 2013.
- Worku AG, Tessema GA, Zeleke AA. Trends and Determinants of Contraceptive Use among Young Married Women (Age 15-24). DHS Working Papers No. 103. Rockville: ICF International; 2014. Epub ahead of print 2014. https://doi.org/10.1371/journal.pone.0116525.
- Haq I, Sakib S, Talukder A. Sociodemographic factors on contraceptive use among ever-married women of reproductive age: evidence from three demographic and health surveys in Bangladesh. Med Sci. 2017;5:31.
- 27. Alabi O, Odimegwu CO, De-Wet N, et al. Does female autonomy affect contraceptive use among women in northern Nigeria? Afr J Reprod Health. 2019;23:92–100.
- Viswan SP, Ravindran TKS, Kandala NB, et al. Sexual autonomy and contraceptive use among women in Nigeria: findings from the demographic and health survey data. Int J Women's Health. 2017;9:581–90.
- Osmani AK, Reyer JA, Osmani AR, et al. Factors influencing contraceptive use among women in Afghanistan: secondary analysis of Afghanistan health survey 2012. Nagoya J Med Sci. 2015;77:551–61.
- Asa SS, Titilayo A, Kupoluyi JA. Assessment of contraceptive use by marriage type among sexually active men in Nigeria. Int Q Commun Health Educ. 2018;38:181–94.
- Thapa S, Mishra V. Mass media exposure among urban youth in Nepal. Asia-Pac Popul J. 2003;18:5–82.
- Ferry E, Ching-Min C, Kurniati A, et al. Determinants of utilization of antenatal care services among adolescent girls and young women in Indonesia. Women Health. 2017;57:614–29.
- Kavitha N. Are young mothers in India deprived of maternal health care services? A comparative study of urban and rural areas. J Health Manag. 2015;17:204–20.
- 34. Dansou J, Adekunle AO, Arowojolu AO. Factors associated with antenatal care services utilisation patterns amongst reproductive age women in Benin Republic: an analysis of 2011/2012 Benin Republic's demographic and health survey data. Niger Postgrad Med J. 2017;24:67–74.
- Woldemicael G, Tenkorang EY. Women's autonomy and maternal healthseeking behaviour in Ethiopia. Matern Child Health J. 2010;14:988–98.
- Choulagai B, Onta S, Subedi N, et al. Barriers to using skilled birth attendants' services in mid- and far-western Nepal: a cross-sectional study. BMC Int Health Hum Rights. 2013;13:49.
- Manyeh AK, Akpakli DE, Kukula V, et al. Socio-demographic determinants of skilled birth attendant at delivery in rural southern Ghana. BMC Res Notes. 2017;10:1–7.
- Sk MIK, Sk MF, Anand A, et al. Utilization of skilled birth attendant in the south Asian region : a comparative study of India, Bangladesh, and Nepal. Eur J Pharm Med Res. 2017;4:305–11.
- Al KGM, Ghosh S, Hossen S, et al. Factors affecting deliveries attended by skilled birth attendants in Bangladesh. Matern Health Neonatol Perinatol. 2017;3:1–9.
- Hazarika I. Factors that determine the use of skilled care during delivery in India: implications for achievement of MDG-5 targets. Matern Child Health J. 2011;15:1381–8.
- Gautam S, Jeong HS. The role of women's autonomy and experience of intimate partner violence as a predictor of maternal healthcare service utilization in Nepal. Int J Environ Res Public Health. 2019;16(5):895. https:// doi.org/10.3390/ijerph16050895.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

